Author's response to reviews

Title: Cystitis due to the use of ketamine as a recreational drug: a case report.

Authors:

    Britt Colebuniders (brittcolebuniders@hotmail.com)
    Peter Van Erps (Peter.VanErps@zna.be)

Version: 2 Date: 17 April 2008

Author's response to reviews:

Response to the reviewers:

Reviewer 1: K L Jansen

- Mysterious pains in the abdominal/pelvic region have long been reported by ketamine addicts, but have not been considered in the peer-reviewed literature. This report contributes to considering possible causes for such pains. # we agree

- There are a few grammatical problems. I suggest careful proofreading by a person who has English as their first language. # The text has been revised by three colleagues (Medical students), who are native English speakers: Elizabeth Concannon, Sarah OShaughnessy, Gearoid OConnor.

Reviewer 2: Anand Patel

- I am happy to recommend acceptance once some spelling errors have been corrected. # The text has been revised by three colleagues (Medical students), who are native English speakers: Elizabeth Concannon, Sarah OShaughnessy, Gearoid OConnor.